

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Sanginabadi, Bahram

# Article

# Does schooling causally impact non-cognitive skills? Evidence from elimination of social security student benefits

Economies

# Provided in Cooperation with:

MDPI - Multidisciplinary Digital Publishing Institute, Basel

*Suggested Citation:* Sanginabadi, Bahram (2020) : Does schooling causally impact non-cognitive skills? Evidence from elimination of social security student benefits, Economies, ISSN 2227-7099, MDPI, Basel, Vol. 8, Iss. 1, pp. 1-27, https://doi.org/10.3390/economies8010005

This Version is available at: https://hdl.handle.net/10419/257055

#### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.



https://creativecommons.org/licenses/by/4.0/

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.







## Article

# Does Schooling Causally Impact Non-Cognitive Skills? Evidence from Elimination of Social Security Student Benefits

## Bahram Sanginabadi

Economics Department, University of Hawaii at Manoa, Honolulu, HI 96822, USA; bahram@hawaii.edu

Received: 25 October 2019; Accepted: 5 January 2020; Published: 14 January 2020



**Abstract:** A limited number of studies have investigated the impacts of education on non-cognitive skills, yet they offer mixed results. A few studies suggest no impact, but others report positive impacts of education on non-cognitive skills. In this paper, we apply the elimination of Social Security student benefits that took place in the United States in 1982 to study the impacts of education on non-cognitive skills, as measured by the Rotter Locus of Control Scale and Rosenberg Self-Esteem Scale. We apply eligibility for aid due to the death of father to avoid endogeneity in our analysis. Our results suggest that non-cognitive skills improve during the college education years, but the causality relationship from college education to non-cognitive skills disappears to a high extent when the prior levels of non-cognitive skills are controlled for.

Keywords: schooling; non-cognitive skills; self-esteem; locus of control

# 1. Introduction

The two well-known measures of non-cognitive skills, the locus of control scale, and the self-esteem scale, have important educational and professional applications. Numerous studies focus on estimating the impacts of non-cognitive skills on important life outcomes, but studies on the impacts of formal schooling on non-cognitive skills remain rare. While researchers have paid attention to the impacts of education on cognitive skills, little attention is given to non-cognitive skills. Focus has been especially centered on crystallized intelligence (Ceci 1991; Murray and Herrnstein 1994; Stelzl et al. 1995; Neal and Johnson 1996; Winship and Korenman 1997; Hansen et al. 2004; Blair et al. 2005; Cascio and Lewis 2006; Cliffordson and Gustafsson 2008; Arcidiacono et al. 2010; Carlsson et al. 2015; Gustafsson 2016; Jonsson et al. 2017).

In addition, the existing studies that investigate the relation between education and non-cognitive skills provide mixed results. Some argue that education affects non-cognitive skills. Others claim that the impacts are significant for younger students, but not for the adolescents and adults. Additionally, it has been discussed that positive impacts of schooling on non-cognitive skills disappear after controlling for the previous levels of non-cognitive skills.

In this research, we apply a shift in Social Security students aid policy that took place in 1982 to study the impacts of an exogenous variation in years of schooling on non-cognitive skills. The reason for this approach is that the endogeneity of education is a well-known issue in studying the impacts of education on wages or intelligence. Applying an exogeneous variation in years of schooling helps to overcome this issue. Hence, we apply a source of variation in schooling that is exogenous to unobservable variables that might influence non-cognitive abilities. We apply a shift in aid policy that affects the school attendance of some students, but not all of them to control for the endogeneity of education.

The study conducted by Heckman et al. (2013) is perhaps the most relevant study to ours. In their research, Heckman and his co-authors study the outcomes of the Perry Preschool Project (1962 to 1967) a randomized trial with a control group designed to provide preschool education to three to four years-old children living in poverty. The project had long-term effects on male income, adult crime rate, and test scores. However, it did not have any significant and lasting impact on IQ. Heckman et al. (2013) argue that the positive effects of the program are due to improvements in the non-cognitive abilities of the children. Additionally, Dee and West (2011) study the non-cognitive returns to class size and find that the reduction in the eighth grade class sizes in Germany is associated with an improvement in non-cognitive skills. Nevertheless, there has been doubt about the impacts of college education on non-cognitive skills, even though the impacts could be strong in younger ages. For example, Bachman et al. (1978) find that college education has little or no impact on self-esteem. Other researchers, such as Kautz et al. (2014), however, argue that non-cognitive skills are malleable in adolescent years and early adulthood and families and years of schooling influence them<sup>1</sup>.

Psychologists have been studying personality traits<sup>2</sup> for a long time when compared with economists. When two influential studies by Heckman and co-authors were published, economists began paying attention to the non-cognitive literature. Nevertheless, the term "non-cognitive" has remained as controversial since then. The first of the two studies by Heckman and Rubinstein (2001) suggest that non-cognitive skills contribute to differences in the earnings of the two groups of GED recipients and high school graduates with similar degrees. They argue that the reason behind the difference in the income of the mentioned groups is that the GED recipients have a shortfall of non-cognitive skills as compared to the high school graduates. Heckman et al. (2006) applied the National Longitudinal Survey of Youth (NLSY) 1979 to study the importance of non-cognitive skills. They apply the two measurements of non-cognitive skills, including Rotter Locus of Control Scale and Rosenberg Self-Esteem Scale, as available in NLSY1979 dataset, and they suggest that non-cognitive abilities are equally as important as cognitive abilities in determining a variety of social and economic outcomes. Heckman et al. (2006) offer two paths through which non-cognitive abilities, including locus of control and self-esteem, could improve earnings. One path is direct effects on productivity in the labor-market and the other is through the indirect effects of non-cognitive abilities on schooling and work experience.

There has been debate in the literature regarding the importance of self-esteem in predicting life outcomes. In research that was conducted by Boden et al. (2007, 2008), the impacts of self-esteem on life outcomes became insignificant after they controlled for 23 covariates in their specifications. However, including that other researchers (Orth and Robins 2014) have criticized many control variables. Another study by Baumeister et al. (2003) argues that self-esteem is not a major predictor of educational and job performance. However, after that, Baumeister and co-authors published their research, many studies found self-esteem as a significant predictor of major life outcomes, as well as educational and job success (Trzesniewski et al. 2006; Orth et al. 2009; Orth et al. 2012; Kuster et al. 2013; Marshall et al. 2014). Additionally, in a review of the literature, Diener (1984) concludes that self-esteem is the most accurate predictor of life satisfaction in the United States, being stronger than physical health, age, education, income, and marital status. In addition, Orth and Robins (2014) argue that self-esteem predicts success and well-being in health outcomes, job market, and relationships. They maintain "given the increasing evidence that self-esteem has important real-world consequences, the topic of self-esteem development is of considerable societal significance" (Orth and Robins 2014, p. 381).

Kautz et al. (2014), particularly argue that non-cognitive skills are more malleable than cognitive skills in adolescent years. Personality traits have been, to a high extent, considered heritable. Twin studies show that 40 to 60 percent of variation in

non-cognitive skills and personality traits is due to genetic causes (Bouchard and Loehlin 2001). Nevertheless, they are under influence of environment, family background, and years of schooling as well.

The previous studies have discussed different ways that years of schooling can affect self-esteem<sup>3</sup>. For example, social comparison at school, which refers to comparing own work with those of the peers, might positively or negatively affect the self-esteem of the students. It is argued that social comparison serves several goals, such as self-presentation, mastery, self-assessment, and self-enhancement (Taylor and Lobel 1989; Butler 1989; Ruble and Frey 1991; Pomerantz et al. 1995). Social comparison boosts the student's self-esteem if they realize that their work is more complete than that of their classmates. However, if the students come to believe that they are less capable than their classmates then social comparison lowers their self-esteem (Pomerantz et al. 1995). Even, the damage to self-esteem of the students could be so serious that they might avoid it altogether. Brickman and Bulman (1977) argue that, despite the fact that social comparison might have a potential for self-enhancement, but it might be avoided because it can damage the self-esteem of own or others. For example, a student who looks at her classmate's work for self-assessment might be disappointed to know that she has accomplished far fewer homework assignments than her classmate. Additionally, a student who shows her finished work to her peers for the purpose of self-enhancement and realizes that they have not completed any of the assignment questions may feel embarrassed by her own superiority (Brickman and Bulman 1977).

Besides social comparison, school performance has been discussed to affect self-esteem. For example, Skaalvik and Hagtvet (1990) find that good performance at school one year leads to a higher self-esteem the year after. Additionally, Rosenberg et al. (1989) find a causality relationship from students' grades to self-esteem. They suggest that having a good performance at school make the students to think of themselves as capable at doing schoolwork, which can boost their self-esteem. Several studies find a correlation between school performance and self-esteem, but they often do not find any causality from self-esteem to school performance, even though they find causality from performance to self-esteem (Rosenberg et al. 1989; Skaalvik and Hagtvet 1990; Bowles 1999; Baumeister et al. 2003). Other studies, such as the one conducted by Rubin et al. (1977), suggest that the correlation between academic performance and self-esteem is based on important underlying factors, such as background and ability.

Another way that the impacts of years of schooling on self-esteem could be explained is through what that psychologists refer to as contingency of self-worth. Crocker and Wolfe (2001) define contingency of self-worth as "a domain or category of outcomes on which a person has staked his or her self-esteem, so that person's view of his or her value or worth depends on perceived successes or failures or adherence to self-standards in that domain" (Crocker and Wolfe 2001, p. 594). They argue that families, institutions, and social groups have implicit or explicit criteria for what is valuable and worthwhile in people. For example, universities value academic achievements. Over time, individuals internalize these criteria and they evaluate their own worth according to them (Crocker and Wolfe 2001). Success in achieving goals that are relevant to one's contingencies of self-worth may boost her self-esteem. For example, students might believe being smart is what makes them worthwhile. They try to get high grades, because it shows that they are smart. Subsequently, getting high grades improves their self-esteem. Additionally, when people believe a higher academic degree is what makes them worthwhile, receiving a higher academic degree might boost their self-esteem. However, people might have motivations that have little or nothing to do with contingency of self-worth. For example, a student might try to get high grades, or someone might try to receive a high academic degree only to facilitate landing a high paying job or to apply for a professional program (Crocker and Wolfe 2001; Crocker et al. 2002).

Locus of control, which is the other non-cognitive skill measurement in this paper, has also been positively associated with labor market outcomes (Heineck and Anger 2010; Cobb-Clark 2015; Caliendo et al. 2015). As an example, Heineck and Anger (2010) find that the external<sup>4</sup> locus of control

<sup>&</sup>lt;sup>3</sup> Self-esteem improves from adolescence to about age 50 to 60 years-old when it peaks. Then, it begins to decline at an accelerating rate until old age. Even though self-esteem is immutable, it is also relatively stable. In addition, someone with a high self-esteem might go through a stage of low self-esteem after a decade and vice versa (Orth and Robins 2014).

<sup>&</sup>lt;sup>4</sup> The meanings of internal and external locus of control are provided in Data and Identification Strategy section.

is associated with a decrease in wages. Caliendo et al. (2015) also argue that individuals with a more internal locus of control scale search harder for jobs when compared to the ones with a more external locus of control scale. Additionally, the individuals with internal locus of control scale have a stronger belief that investments in job search would pay off later. Additionally, individuals with external locus of control scale have lower reservation wages<sup>5</sup> (Caliendo et al. 2015).

It has been discussed in the literature that one of the most influential interventions on locus of control is achieving a higher social status (Wolfle and Robertshaw 1982). For example, Harvey (1971) maintains that upward mobility is associated with a change in locus of control toward internality. That means the individuals that experience an upward shift in their social status are likely to experience a change in their locus of control scale from external to internal levels. Particularly, education can create a path toward a higher social status and has the potential to affect locus of control toward a more internal level. However, it is worth noting that studies, such as the one by Wolfle and Robertshaw (1982), argue that the causal relationship between social status mobility and locus of control is a two-way street. Wolfle and Robertshaw (1982) emphasize that it is not surprising that new experiences, environments, and social conditions can lead to shifts in personality characteristics, because those who experience a shift in their status might have a high degree of internal control in the first place. Therefore, part of the correlation between upward social status mobility and personality characteristics can be attributed to the same personality characteristics that affect social mobility. In a review of the literature, Pascarella and Terenzini (1991) provide ample evidence that the locus of control increases during college education years (i.e., it becomes more internalized). However, Wolfle and Robertshaw (1982) suspect that the college students not only acquire higher education, but they are already internally oriented. Wolfle and Robertshaw (1982) argue that the strongest explanatory variable of the locus of control among high school students is ability such that those with higher abilities have more internalized locus of control. However, Wolfle and List (2004) results show that college education affects the locus of control of the students and shift it from external toward internal levels, even after controlling for ability, background, and previous levels of locus of control.

Non-cognitive skills<sup>6</sup> are not limited to the Rosenberg Self-Esteem Scale<sup>7</sup> and Rotter Locus of Control Scale<sup>8</sup>. They also include other skills. In this paper, we apply the Rotter Locus of Control Scale and Rosenberg Self-Esteem Scale for the exact same reason that Heckman and his co-authors use them. As they point out, "we choose these measures because of their availability in the NLSY79" (Heckman et al. 2006, p. 429). Ideally, using a wider range of psychological measurements is preferable. Nevertheless, the Rotter Locus of Control Scale and Rosenberg Self-Esteem Scale are two essential measurements of non-cognitive skills with significant educational and professional outcomes. Heckman et al. (2006) and other researchers, such as Waddell (2006) and Deke and Haimson (2006), argue that the Rotter Locus of Control Scale and Rosenberg Self-Esteem Scale are associated with long term labor market outcomes.

Numerous studies point out the importance of non-cognitive skills on labor market outcomes (e.g., Gintis 1971; Heckman and Rubinstein 2001; Bowles et al. 2001; Kuhn and Weinberger 2005; Heckman et al. 2006; Waddell 2006; Fortin 2008; Deke and Haimson 2006; Heineck and Anger 2010; Dee and West 2011; Drago 2011; de Araujo and Lagos 2013; de Araujo and Lagos 2013; Cobb-Clark 2015). A reasonable clue that shows the importance of non-cognitive skills in the job market comes from the

<sup>&</sup>lt;sup>5</sup> Reservation wage is defined as the lowest wage that an employee accepts for a particular job.

<sup>&</sup>lt;sup>6</sup> Note that psychologists and economists have used a variety of terminologies to describe non-cognitive skills. Other terms such as non-cognitive abilities, character skills, soft skills, socio-emotional skills, and personality traits have been used to refer to the same range of abilities or skills (Heckman and Kautz 2013; Garcia 2016). A well-known and widely accepted classification of personality traits (non-cognitive skills) is the "Big Five", which includes agreeableness, extraversion, neuroticism, conscientiousness, and openness to experience.

<sup>&</sup>lt;sup>7</sup> Rosenberg Self-Esteem Scale measures an individual's subjective belief about her/his own worth. More details on how this scale is measured is provided in Appendix A.

<sup>&</sup>lt;sup>8</sup> Rotter Locus of Control Scale measure the extend which individuals believe have control over their lives. More details on how this scale is measured is provided in Appendix B.

behavior of the employers. In a survey of 400 employers in the United States (U.S.), Casner-Lotto and Barrington (2006) find that the top four skills that employers look for in applicants are oral communication, teamwork, work ethic and professionalism, and problem solving and critical thinking. Writing abilities, mathematics knowledge, and other cognitively demanding skills ranked lower than the mentioned soft skills. Self-esteem, which is one of the non-cognitive measurements that we use in this paper, has been associated with teamwork and higher levels of perseverance, two top soft skills that employers seek in job candidates (Murnane et al. 2001). Another finding of Murnane et al. (2001) is that there is a positive association between self-esteem and earnings 10 years later. Additionally, Gintis (1971) argues that, if non-cognitive abilities are omitted in return to schooling specifications, then the estimated impacts of education on earnings would be biased due to the impacts of non-cognitive abilities on earnings.

Besides what is argued above, this study can help with an old debate in the literature: human capital versus signaling views. Positive impacts of education on non-cognitive skills is in favor of human capital view rather than signaling view. However, insignificant impacts of education on non-cognitive skills support the signaling view. According to the signaling view by Spence (1978), a school degree reveals the ability of a prospective employee to a potential employer by sending her a signal. Based on this view, which Michael Spence and other scholars have developed, credentials reveal the unobserved ability of the employees to the employers, which aids the employers to distinguish low and high ability workers from each other. After the employee enters the job market and starts working, the role of ability in determining the wages increases, because ability is more observable to the employer. Over time, the informational role of credentials in determining wages decreases because the employers can observe the employee's abilities. According to this view, people with higher levels of productivity choose higher levels of education to signal their ability to potential employers. Arcidiacono et al. (2010) apply the Armed Forces Qualification Test (AFQT) as a measure of ability and find that the ability of high school graduates is revealed gradually and over time to the job market, but the ability of college graduates is almost perfectly observed in the job market. The finding of this study suggest that a college degree might serve a signaling purpose. According to the human capital view that Gary Becker and other scholars developed, investment in education enhances productivity and, therefore, income. Becker argues that "Education, training, and health are the most important investments in human capital (Becker 1992, p. 1)". The supporters of human capital view argue that credentials do not serve employees merely due to their informational value in sending signals to the employers, but people improve their human capital by attending school (Becker 1964; Ben-Porath 1967).

As mentioned before, in this paper, we apply a shift in aid policy that took place in 1982 to study the impacts of an exogenous variation in years of schooling on non-cognitive abilities, including self-esteem and locus of control. Our results suggest that the impact of schooling on the Rosenberg Self-Esteem Scale and the Rotter Locus of Control Scale is such that it improves self-esteem and increases the sense of control over life (internal locus of control) when the previous levels of non-cognitive skills are not controlled for, but when those variables are included as the control variable then the results are hardly statistically significant.

The rest of this paper is organized, as follows. Section 2 explains the data and identification strategy. Section 3 outlines the empirical design of the paper. Section 4 provides the results and Section 5 presents the conclusion of the study.

## 2. Data and Identification Strategy

#### 2.1. Data

NLSY79 is a nationally representative sample of 12,686 young American men and women aged between 14 and 22 years when they were first surveyed in 1979. After 1979, the data have been collected

in most of the years until 2016. Table 1 represents the number and percentage of each gender, and ethnic group is represented in NLSY79.

|                      | Number | Percentage |
|----------------------|--------|------------|
| Total                | 12,686 | 100        |
| Hispanic             | 2002   | 15.78      |
| Blacks               | 3174   | 25.02      |
| Non-Hispanic & Black | 7510   | 59.19      |
| Males                | 6403   | 50.47      |
| Females              | 6283   | 49.53      |

 Table 1. Characteristics of NLSY79 dataset.

We use the Rosenberg Self-Esteem Scale, which was designed by Rosenberg (1965), as one of the measurements of non-cognitive ability. NLSY79 provides the Rosenberg Self-Esteem Scale in the 1980, 1987, and 2006 interviews. In this scale, the respondents are asked to express their opinions regarding 10 statements of self-approval and disapproval. The respondents are asked to pick one of the options among strongly agree, agree, disagree, or strongly disagree. An example of a Rosenberg Self-Esteem Scale statement is "I am a person of worth". A higher score indicates higher self-esteem. Please see Appendix A for more details regarding the Rosenberg Self-Esteem Scale, including the questions that are asked from the participants, the number of participants, and the scores.

The Rotter Locus of Control Scale designed by Rotter (1966) is the other variable that we use as a measurement of non-cognitive skills. The NLSY79 provides Rotter Locus of Control Scale in 1979 and 2014. This variable measures the extent that individuals believe that they have control over their lives, which can be explained as internal control versus external control. Internal control is about the extent by which individuals believe that they have control over their lives through self-motivation or self-determination, while external control is about the extent that they believe the environment (that is, chance, fate, luck, etc.) controls them. Higher scores indicate that the individual is more external. However, we reverse the sign of locus of control scores in this paper, so the results are consistent with those of the self-esteem coefficients.

Note that we standardize self-esteem and locus of control scales, such that their means equal to zero and their standard deviations set to one. This enables us to interpret the impacts of education on the mentioned scales in percentage points.

#### 2.2. Identification Strategy

Studying the impacts of education on abilities is a challenge, because an improvement in abilities could also improve education. In addition, unobserved variables that might affect both education and abilities contribute to the endogeneity of education. Therefore, the estimated impacts of education on ability could be biased due to the endogeneity of education. Nevertheless, exogenous variation in years of schooling helps to overcome the problem of endogeneity of education. We need a source of variation in schooling that is exogenous to unobservable variables that might influence non-cognitive abilities to overcome this issue. In this paper, we apply a shift in aid policy that affects the school attendance of some students but not all of them to study the impacts of schooling on non-cognitive abilities. The elimination of a Social Security student benefit program in 1982 is the policy that we apply.

From 1965 to 1982, the Social Security Administration provided aid for the 18 to 22 years old children of Social Security beneficiaries, including the children of deceased, disabled, and retired parents. Prior to 1965, aid was provided to the children of Social Security beneficiaries only up to age 18. From 1965 to 1982, the Social Security Administration covered students up to age 22 if they were full time college students. According to the data from National Center for Education Statistics, the program covered 700,000 college students at its peak. From 1977 to 1982, the number of covered students dropped moderately to 600,000, due to a national drop in college enrolment rates in the U.S.

(Dynarski 2003). In 1981, Congress voted to eliminate the Social Security student benefit program. Students who were not enrolled in college by May 1982 were no longer eligible to receive aid. Moreover, payments to the ones who were enrolled in college were significantly reduced.

We apply eligibility for aid due to death of the parent in this paper since parental retirement and disability could be endogenous due to availability of student benefits. Following Dynarski (2003), we focus on the students with deceased fathers since 90 percent of the students were eligible for the program from that cause. Note that the focus of the paper is estimating the impacts of aid eligibility on education and non-cognitive abilities, rather than estimating the impacts of aid receipt on the mentioned variables. This approach enables us to unbiasedly predict the impact of eliminating the aid policy, because the policy makers control the offer of the aid but not its take up.

Table 2 shows the mean and standard error of the education level of the students with deceased and not deceased father before and after the policy change. Schooling by age 23 is the number of years of schooling that a person has completed by age 23. The variable complete college by age 23 is a binary variable that sets one if an individual completed any years of college by age 23. Otherwise, it is equal to zero. Complete college by age 28 equals to one if a person completed any years of college before age 28. For each variable listed in Table 2, the number of observations is provided underneath the mean and standard error. Standard errors are in parentheses. The mean of all the variables in the table show that children of deceased fathers are better off in terms of education before the change in policy in 1982. For example, their years of schooling by age 23 that has been 13 and larger than that of other students has decreased to 12.7, while years of schooling of the regular students has increased. The last column of Table 2 shows the difference-in-differences impact of eligibility for Social Security student benefits on educational outcomes. The positive means in the last column show the positive impact of the student eligibility for the Social Security student aid program on education outcomes. Standard errors in this column are clustered by household, due to the presence of siblings in some families. Table 2 provides some suggestive evidence for the difference-in-differences findings and the results in this table will be tested in the next sections of the paper.

|                     | High Scho<br>1979- |          | High Scho<br>1982- |          |                |
|---------------------|--------------------|----------|--------------------|----------|----------------|
|                     | Father Not         | Father   | Father Not         | Father   | Difference-in- |
|                     | Deceased           | Deceased | Deceased           | Deceased | Differences    |
| schooling by age 23 | 12.839             | 13       | 12.852             | 12.702   | 0.309          |
|                     | (0.050)            | (0.090)  | (0.095)            | (0.175)  | (0.215)        |
| Number of obs.      | 699                | 247      | 210                | 74       | 1230           |
| complete college    | 0.372              | 0.425    | 0.394              | 0.358    | 0.089          |
| by age 23           | (0.018)            | (0.031)  | (0.033)            | (0.054)  | (0.071)        |
| Number of obs.      | 716                | 256      | 218                | 81       | 1271           |
| complete college    | 0.461              | 0.505    | 0.552              | 0.419    | 0.177          |
| by age 28           | (0.019)            | (0.031)  | (0.034)            | (0.055)  | (0.074)        |
| Number of obs.      | 722                | 257      | 219                | 81       | 1279           |

| Table 2. NLS | r Summary | Statistics. |
|--------------|-----------|-------------|
|--------------|-----------|-------------|

Note: Schooling by age 23 is the number of years of schooling that an individual has completed by age 23. Complete college by age 23 is a binary variable that equals to one if an individual has completed any years of college by age 23. Otherwise, it is equal to zero. Complete college by age 28 equals to one if a person completed any years of college by age 28. Otherwise, it is equal to zero.

#### 3. Empirical Design

We apply the difference-in-differences approach to study the impacts of eligibility for the Social Security benefits on educational outcomes. To do so, we estimate the following standard difference-in-differences model:

Schooling<sub>i</sub> = 
$$\beta_1 + \beta_2(father \, deceased_i \times before_i) + \beta_3 father \, deceased_i + \beta_4 before_i + \beta_5 X_i + \epsilon_i$$
 (1)

where *Schooling*<sub>i</sub> stands for educational outcomes for individual *i*. In the empirical results section of the paper, when we show the results, we clarify what exactly *Schooling*<sub>i</sub> stands for in each estimated regression. *Father deceased* is a binary variable that is equal to one for those who are potentially eligible for Social Security student benefits, due to death of their fathers. Also *before*<sub>i</sub>, is a binary variable that shows that individual *i* graduated from high school before the elimination of the program. *before*<sub>i</sub> equals to one if individual *i* is a high school senior in 1979, 1980, or 1981. The NLSY79 dataset that we use starts in 1979. Therefore, we do not have data prior to 1979. *before*<sub>i</sub> is zero if individual *i* is a high school senior in 1982 or 1983 (i.e., after elimination of the program). The reason that high school seniors after 1983 are not included in our regressions is that they age out of the 18–22 years old eligibility for the program. *X*<sub>i</sub> represents a vector of family background control variables and the birth year and region fixed effects.

We apply instrumental variables specifications to address this issue since the literature discusses that schooling could be endogenous due to unobserved innate ability. In a regular least square specification that aims to estimate the impacts of education on labor market outcomes, innate ability is in the error term and it might be correlated with the dependent variable.

We are interested in understanding the impacts of years of schooling on non-cognitive skills (i.e., Rosenberg Self-Esteem Scale and Rotter Locus of Control Scale). This relationship can be specified, as follows:

$$Y_i = \delta_1 + \delta_2 Schooling_i + \delta_3 father \ deceased_i + \delta_4 before_i + \delta_5 X_i + \epsilon_i \tag{2}$$

where  $Y_i$  is the outcome variable for individual *i*. The outcome variable in our regressions is either the Rosenberg Self-Esteem Scale or Rotter Locus of Control Scale. The rest of the variables are introduced before.

We apply the interaction between *father deceased* and *before* as an instrument in the following model, which serves as the first stage of our two-stage least-square estimations specification for Equation (2):

$$Schooling_i = \theta_1 + \theta_2(father \, deceased_i \times before_i) + \theta_3 father \, deceased_i + \theta_4 before_i + \theta_5 X_i + \omega_i$$
(3)

where all the variables are introduced in Equations (1) and (2). The interaction of *father deceased* and *before* is an exogenous variable that affects years of education, but it does not directly affect noncognitive abilities.

#### 4. Empirical Results

In this section, Table 3 provides the estimates of the difference-in-differences model (i.e., Equation (1)) and Tables 4–7 present the estimates of the least-square and the two-stage least-square models (i.e., Equations (2) and (3)). The goal in Table 3 is to estimate the impacts of aid eligibility on schooling. This helps us to realize whether the elimination of the Social Security student benefits causes any variations in schooling outcomes. In Tables 4 and 5, the least-square methodology is applied to estimate the impacts of schooling on non-cognitive abilities. As discussed before, the results from least-square estimations could be biased due to the possible endogeneity of schooling. Therefore, in Tables 6 and 7, the two-stage least-square estimations that apply instrumental variables to control for the endogeneity of schooling are provided. The two-stage least-square regressions estimate the impacts of variation in years of schooling on non-cognitive abilities.

|                                 | Schooling | by Age 23 |         | College by<br>e 23 |           | College by<br>je 28 |
|---------------------------------|-----------|-----------|---------|--------------------|-----------|---------------------|
| father deceased $\times$ before | 0.309     | 0.491 *   | 0.089   | 0.238 **           | 0.177 **  | 0.293 ***           |
| latilei deceased × belore       | (0.215)   | (0.259)   | (0.071) | (0.085)            | (0.074)   | (0.088)             |
| father deceased                 | -0.149    | -0.348    | -0.036  | -0.180 **          | -0.132 ** | -0.257 ***          |
| lattiel deceased                | (0.199)   | (0.235)   | (0.063) | (0.075)            | (0.064)   | (0.078)             |
| before                          | -0.012    | 0.311     | -0.021  | -0.503 **          | -0.091 ** | -0.118              |
|                                 | (0.107)   | (0.884)   | (0.037) | (0.244)            | (0.038)   | (0.295)             |
| family income                   |           | Y         |         | Y                  |           | Y                   |
| black                           |           | Y         |         | Y                  |           | Y                   |
| Hispanic                        |           | Y         |         | Y                  |           | Y                   |
| father attended college         |           | Y         |         | Y                  |           | Y                   |
| mother attended college         |           | Y         |         | Y                  |           | Y                   |
| single parent family            |           | Y         |         | Y                  |           | Y                   |
| family size                     |           | Y         |         | Y                  |           | Y                   |
| age88                           |           | Y         |         | Y                  |           | Y                   |
| female                          |           | Y         |         | Y                  |           | Y                   |
| age^2                           |           | Y         |         | Y                  |           | Y                   |
| married                         |           | Y         |         | Y                  |           | Y                   |
| nosmsa                          |           | Y         |         | Y                  |           | Y                   |
| centrsmsa                       |           | Y         |         | Y                  |           | Y                   |
| Zasvab                          |           | Y         |         | Y                  |           | Y                   |
| Year of birth fixed effect      |           | Y         |         | Y                  |           | Y                   |
| Region fixed effect             |           | Y         |         | Y                  |           | Y                   |
| All covariates × before         |           | Y         |         | Y                  |           | Y                   |
| Number of obs.                  | 1230      | 875       | 1279    | 884                | 1279      | 884                 |
| R-squared                       | 0.003     | 0.189     | 0.046   | 0.171              | 0.005     | 0.177               |

Table 3. Impact of eligibility for Social Security student benefits on education.

Note: Schooling by age 23 is the number of years of schooling that an individual has completed by age 23. Complete college by age 23 is a binary variable that equals to one if an individual has completed any years of college by age 23. Otherwise, it is equal to zero. Complete college by age 28 equals to one if a person completed any years of college by age 28. Otherwise, it is equal to zero. \*: Significant at 10%; \*\*: Significant at 5%; \*\*\*: Significant at 1%.

| Dependent<br>Variable | (1)                  | N. of<br>obs. | R2    | (2)                  | N. of<br>obs. | R2    | (3)                  | N. of<br>obs. | R2    | (4)                  | N. of<br>obs. | R2    | (5)                  | N. of<br>obs. | R2    | (6)                  | N. of<br>obs. | R2    |
|-----------------------|----------------------|---------------|-------|----------------------|---------------|-------|----------------------|---------------|-------|----------------------|---------------|-------|----------------------|---------------|-------|----------------------|---------------|-------|
| Z. Rosenberg<br>1987  | 0.135 ***<br>(0.019) | 1173          | 0.045 | 0.127 ***<br>(0.023) | 869           | 0.069 | 0.127 ***<br>(0.023) | 869           | 0.071 | 0.136 ***<br>(0.023) | 869           | 0.082 | 0.127 ***<br>(0.023) | 858           | 0.090 | 0.140 ***<br>(0.023) | 858           | 0.140 |
| Z. Rosenberg<br>2006  | 0.064 ***<br>(0.022) | 883           | 0.011 | 0.050 *<br>(0.027)   | 640           | 0.038 | 0.050 *<br>(0.027)   | 640           | 0.039 | 0.054 *<br>(0.028)   | 640           | 0.042 | 0.050 *<br>(0.027)   | 631           | 0.071 | 0.048<br>(0.029)     | 631           | 0.118 |
| Z. Rotter 2014        | 0.095 ***<br>(0.024) | 760           | 0.021 | 0.072 **<br>(0.031)  | 557           | 0.041 | 0.073 **<br>(0.031)  | 557           | 0.041 | 0.079 **<br>(0.031)  | 557           | 0.050 | 0.076 *<br>(0.031)   | 549           | 0.060 | 0.075 **<br>(0.031)  | 549           | 0.146 |

Table 4. Impact of education on non-cognitive skills—least-square estimations. Independent variable of interest: schooling by age 23.

Note: The upper numbers in columns (1) to (6) show the coefficient of schooling by age 23. Column (1) include no control variables. Column (2) includes father deceased, before, family income, black, Hispanic, father attended college, mother attended college, single parent family, family size, age88, and female. Column (3) includes all the covariates in column (2) and age powered by 2. Column (4) includes all the covariates in column (3) and married, nosmsa, centrsmsa, Zasvab. Column (5) includes all the covariates in column (2) and ge powered by 2. Column (6) includes all of the covariates in column (5) and the interaction of all covariates and *before*. Family income is the income of the family that individual *i* has grown up there. Black is a binary variable that set to one if individual *i* is black and it is zero otherwise. Hispanic is one if individual *i* is Hispanic and it is zero otherwise. Father and mother attended college are binary variables that show parents of individual *i* have attended college. Single parent is a binary variable that sets to one if individual *i* have attended college. Single parent is a binary variable that sets to one if individual *i* have attended college. Single parent is a binary variable that sets to one if individual *i* have attended college. Single parent is a binary variable that sets to one if individual *i* have attended college. Single parent is a binary variable that sets to one if individual *i* have attended college. Single parent is a binary variable that sets to one if individual *i* have attended college. Single parent is a binary variable that sets to one if individual *i* have attended college. Single parent is a binary variable that sets to one if individual *i* have attended college. Single parent is a binary variable that sets to one if individual *i* is a female and it is zero otherwise. Age squared is age powered by two. Married, is a binary variable and shows whether a person is married or single. Nonsmas sets to one if individual *i* lives in the c

| Dependent<br>Variable | (1)                  | N. of<br>obs. | R2    | (2)                  | N. of<br>obs. | R2    | (3)                  | N. of<br>obs. | R2    | (4)                  | N. of<br>obs. | R2    | (5)                  | N. of<br>obs. | R2    | (6)                  | N. of<br>obs. | R2    |
|-----------------------|----------------------|---------------|-------|----------------------|---------------|-------|----------------------|---------------|-------|----------------------|---------------|-------|----------------------|---------------|-------|----------------------|---------------|-------|
| Z. Rosenberg<br>1987  | 0.381 ***<br>(0.054) | 1188          | 0.045 | 0.362 ***<br>(0.064) | 876           | 0.070 | 0.362 ***<br>(0.064) | 876           | 0.070 | 0.362 ***<br>(0.064) | 876           | 0.072 | 0.386 ***<br>(0.065) | 876           | 0.083 | 0.366 ***<br>(0.065) | 865           | 0.092 |
| Z. Rosenberg<br>2006  | 0.169 ***<br>(0.058) | 898           | 0.010 | 0.160 **<br>(0.072)  | 646           | 0.038 | 0.163 **<br>(0.072)  | 646           | 0.040 | 0.170 **<br>(0.073)  | 646           | 0.042 | 0.154 **<br>(0.073)  | 637           | 0.072 | 0.143 *<br>(0.078)   | 637           | 0.113 |
| Z. Rotter 2014        | 0.306 ***<br>(0.066) | 774           | 0.027 | 0.324 ***<br>(0.083) | 562           | 0.056 | 0.326 ***<br>(0.083) | 562           | 0.057 | 0.339 ***<br>(0.083) | 562           | 0.065 | 0.329 ***<br>(0.084) | 554           | 0.075 | 0.315 ***<br>(0.086) | 554           | 0.156 |

Table 5. Impact of education on non-cognitive skills-least-square estimations. Independent variable of interest: complete college by age 23.

Note: The upper numbers in columns (1) to (6) show the coefficient of complete college by age 23. Column (1) include no control variables. Column (2) includes father deceased, before, family income, black, Hispanic, father attended college, mother attended college, single parent family, family size, age88, and female. Column (3) includes all the covariates in column (2) and age powered by 2. Column (4) includes all the covariates in column (3) and married, nosmsa, centrsmsa, Zasvab. Column (5) includes all the covariates in column (4) and year of birth and region fixed effect. Column (6) includes all the covariates in column (5) and the interaction of all covariates and *before*. Family income is the income of the family that individual *i* has grown up there. Black is a binary variable that set to one if individual *i* is black and it is zero otherwise. Hispanic is one if individual *i* is Hispanic and it is zero otherwise. Father and mother attended college are binary variables that show parents of individual *i* have attended college. Single parent is a binary variable that equals to one if individual *i* is a female and it is zero otherwise. Age squared is age powered by two. Married, is a binary variable and shows whether a person is married or single. Nonsmas aests to one if individual *i* does not the four regions in the United States: Northeast, the Midwest, the South, and the Wes All covariates × before t. is all covariates multiplied by *before*. The numbers in parentheses represent standard errors. N. of obs. represents number of observations. R2 stands for R squared. All regressions are clustered at the household level. \*: Significant at 10%; \*\*: Significant at 5%; \*\*\*: Significant at 1%.

**Table 6.** Impact of education on non-cognitive skills—two-stage least-square estimations. Instrumented variable: schooling by age 23.

| Independent Variable                                 | Z. Rosenberg 1987 | Z. Rosenberg 2006 | Z. Rotter 2014 |
|--|-------------------|-------------------|----------------|
|  | 0.269 **          | 0.245 **          | 0.274 *        |
| Coefficient of schooling by age 23                   | (0.105)           | (0.100)           | (0.158)        |
| Number of obs.                                       | 896               | 667               | 576            |
| Test of weak instruments: first stage F              | 11.933            | 11.297            | 14.960         |
| Previous non-cognitive measure as a control variable | NO                | NO                | NO             |
|  | 0.115             | 0.189 *           | 0.262          |
| Coefficient of schooling by age 23                   | (0.114)           | (0.110)           | (0.165)        |
| Number of obs.                                       | 895               | 666               | 573            |
| Test of weak instruments: first stage F              | 9.435             | 9.350             | 14.143         |
| Previous non-cognitive measure as a control variable | Z. Rosenberg 1980 | Z. Rosenberg 1980 | Z. Rotter 1979 |

Instrumented: schooling by age 23. Instrument: interaction of *father deceased* and *before*. Test of weak instrument Ho: Instruments are weak. Control variables: family income, age, and standardized ASVAB score. All regressions are clustered at the household level. The numbers in parentheses represent standard errors. \*: Significant at 10%; \*\*: Significant at 5%; \*\*\*: Significant at 1%.

 Table 7. Impact of education on non-cognitive skills—two-stage least-square estimations. Instrumented variable: complete any years of college by age 23.

| Independent Variable                                 | Z. Rosenberg 1987 | Z. Rosenberg 2006 | Z. Rotter 2014 |
|--|-------------------|-------------------|----------------|
| Coefficient of complete college by age 23            | 0.918 **          | 0.907 **          | 0.928 *        |
| Coefficient of complete conege by age 25             | (0.378)           | (0.360)           | (0.489)        |
| Number of obs.                                       | 905               | 676               | 581            |
| Test of weak instruments: first stage F              | 9.761             | 9.308             | 8.808          |
| Previous non-cognitive measure as a control variable | NO                | NO                | NO             |
| Coefficient of complete college by age 23            | 0.351             | 0.700 *           | 0.850 *        |
| Coefficient of complete conege by age 25             | (0.426)           | (0.401)           | (0.471)        |
| Number of obs.                                       | 904               | 675               | 562            |
| Test of weak instruments: first stage F              | 7.023             | 7.142             | 11.746         |
| Previous non-cognitive measure as a control variable | Z. Rosenberg 1980 | Z. Rosenberg 1980 | Z. Rotter 1979 |

Instrumented: complete any years of college by age 23. Instrument: interaction of *father deceased* and *before*. Test of weak instrument Ho: Instruments are weak. Control variables: family income, age, and standardized ASVAB score. All regressions are clustered at the household level. The numbers in parentheses represent standard errors. \*: Significant at 10%; \*\*: Significant at 5%; \*\*\*: Significant at 1%.

In Table 3, the results for the main difference-in-differences model are estimated with either no control variables or a full set of control variables. Note that Appendix C provides estimations of the difference-in-differences model with four different subsets of control variables. In Table 3, age squared is age powered by two and shows the sensitivity of the results to the functional form of the regression taken by age. Additionally, Nonsmsa is a binary variable that sets to one if individual *i* does not live in a city and it is zero otherwise. Centrsmsa is a binary variable that equals one if individual *i* lives in the central city. Zasvab is the standardized ASVAB score of individual *i*, The ASVAB includes 10 tests that measure skill and knowledge in general science, arithmetic reasoning, paragraph comprehension, word knowledge, coding speed, numerical operations, mathematics knowledge, auto and shop information, electronics information, and mechanical comprehension. A standardized ASVAB score has been added to our covariates to control for knowledge and cognitive abilities of the participants. The region of birth refers to one of the four regions in the United States: Northeast, the Midwest, the South, and the West. 'All covariates × before' is the interaction of all covariates with *before*. This controls for heterogeneity across eligibility status and time. For example, this interaction term absorbs the changes in economy that might affect family income and, therefore, school attendance of the youth.

Table 3 represents the impact of eligibility for student benefits on years of schooling by age 23, complete college by age 23, and complete college by age 28. Note that schooling is not a binary variable, but it is the actual years of schooling. However, complete college by age 23 and 28 are the binary variables. The results do not show the significant impacts of eligibility for student benefits on years of schooling by age 23 and completing any years of college by age 23 when no control variable is included in the regressions. However, in most cases, we see positive and significant impacts when the control

variables are included. These results could be interpreted as positive impacts of the student benefits program on education outcomes. However, these findings should be interpreted with caution because, as mentioned above, the impacts are not significant when the control variables are not included. Another reason is that NLSY79 dataset does not include data prior to 1979. This might also affect the findings that are presented in Table 3. Please see Appendix C for more details.

Family income is the income of the family that individual *i* has grown up there. Black is a binary variable that set to one if individual *i* is black and it is zero otherwise. Hispanic is one if individual *i* is Hispanic and it is zero otherwise. Father and mother attended college are the binary variables that show parents of individual *i* have attended college. Single parent is a binary variable that equals to one if individual *i* has grown up in a single parent household and it equals zero otherwise. Family size refers to actual number of the family members. Female is a binary variable that sets to one if individual *i* is a female and it is zero otherwise. Age squared is age powered by two. Married, is a binary variable and it shows whether a person is married or single. Nonsmsa sets to one if individual *i* does not live in a city, and it is zero otherwise. Centrsmsa equals to one if individual *i* lives in the central city. Zasvab is the standardized ASVAB score of individual *i*, The region of birth refers to one of the four regions in the United States: Northeast, the Midwest, the South, and the West. All covariates × before is all covariates multiplied by *before*.

The upper numbers in each column show the coefficient of the independent variable provided in the first column (i.e., column in the left) of the table. The numbers in parentheses represent the standard errors. Y means that the associated covariate is included in the estimation of the equation. All of th regressions are clustered at the household level. \*: Significant at 10%; \*\*: Significant at 5%; \*\*\*: Significant at 1%.

As mentioned before, the goal of this paper is to understand the impacts of schooling on non-cognitive abilities, rather than the impacts of aid on schooling. Dynarski (2003) studied the impacts of variation in aid on years of schooling and college attendance. Here, we not only replicated her results and found similar impacts of aid on education, but we provided more details on the impacts of aid eligibility on schooling. Our findings to this point give us confidence that we have estimated the impacts of eligibility for Social Security student benefits on schooling correctly. We apply this variation in our two-stage least-square estimations specifications to estimate the impacts of the years of education on non-cognitive abilities. However, we have presented the results of least-square estimations before discussing the two-stage least-square estimations results.

Table 4 shows the results of the least-square estimations of Equation (2). The upper numbers in each cell under columns (1) to (6) show the coefficient of schooling by age 23. The numbers in parentheses represent the standard errors. Column (1) is estimated without any control variables. The covariates in column (2) include family Income, black, Hispanic, father attended college, mother attended college, single parent, family size, age, and female. The covariates in column (3) include all of those in column (2) and age powered by 2. In column (4), all of the covariates of column (3) and four new covariates, including married, nosmsa, centrsmsa, and standardized ASVAB test scores, are added to the regressions. In column (5), all of the covariates in column (4) and year of birth and region of birth fixed effects are added. In column (6), all covariates in column (5) and the interaction of all of covariates with *before* are included.

The first row in the table shows the impacts of schooling by age 23 on the standardized Rosenberg Self-Esteem Scale that is measured in 1987. Note that, in 1987, the youngest participants in NLSY 1979 dataset are 23 years old. The results show that the impact of years of schooling by age 23 on the standardized Rosenberg Self-Esteem Scale that was measured in 1987 is positive and relatively stable in columns (1) to (6). The impacts vary from 12.7 to 14.0 percentage points.

The second row shows the impact of years of schooling completed by age 23 on the standardized Rosenberg Self-Esteem Scale measured in 2006. In 2006, the youngest participant in the NLSY dataset is 42 years old. Hence, other than some rare exceptions, everybody should be out of school by 2006. Here, the coefficients are smaller when compared to those in the first row and we still see some

positive impacts of years of schooling on the Rosenberg Self-Esteem Scale measured in 2006 when the participants ages range from 42 to 49. Nevertheless, note that the coefficients are only statistically significant at 10 percent level in columns (1) to (5) and they are not statistically significant in column (6).

The last row in Table 4 represents the impacts of schooling by age 23 on the Standardized Rotter Locus of Control Scale that was measured in 2014. As can be seen, the coefficients are positive, which suggest that more years of schooling is associated with an increase in the sense of control over life (positive impacts on internal locus of control and negative impacts on external locus of control).

Table 5 represents the impacts of complete any years of college by age 23 on the standardized Rosenberg Self-Esteem Scale that was measured in 1987 and 2006 and standardized Rotter Locus of Control Scale measured in 2014. The results show the positive impacts of complete any years of college by age 23 on non-cognitive abilities. Note that results from least square regressions could be biased due to the endogeneity of education.

Tables 6 and 7 show the two-stage least-square estimations of Equation (3). We have used the interaction of *father deceased* and *before* as an instrument. This instrument has good explanatory power in the first stage. Note that using the interactions between *father deceased* and *after* as an instrument produces exact same results with the same sign as that of the interaction of *father deceased* and *before*. In Table 6 the instrumented variable is schooling by age 23 and in Table 7 the instrumented variable is completed any years of college by age 23. Tables A13–A18 in Appendix D provide the results of the two-stage least-square estimations with different subsets of control variables.

The dependent variable in Table 6 is the standardized Rosenberg Self-Esteem Scale that was measured in 1987. The results are presented in two rows. In the first row, the previous levels of non-cognitive abilities are not controlled for, but they are controlled for in the second row.

Perhaps the most important aspect of the results in Tables 6 and 7 is that education has a significant impact on non-cognitive skills when we do not control for the previous levels of non-cognitive skill, but we hardly see any significant impact when the prior levels of non-cognitive skills are controlled for. In Tables 6 and 7, when a previous level of self-esteem is not included as a control variable, we can see a significant and positive impact of education on self-esteem and when a previous measure of self-esteem is included as a control variable, the impacts are still positive, but they either are not statistically significant or they are only significant at the 10 percent level. The impact of education on locus of control is positive and statistically significant at the 10 percent level when a previous level of locus of control is not controlled for, but when it is controlled for then the impact is not statistically significant.

#### 5. Conclusions

Understanding the true impacts of college education on non-cognitive skills remains a highly crucial question in the literature, as the previous studies have not paid enough attention to this issue, despite the fact that they have highlighted the importance of non-cognitive skills. Multiple studies argue that non-cognitive skills are highly important in daily life and they have crucial life and professional outcomes. For example, Diener (1984) finds self-esteem to be the strongest predictor of life satisfaction in the United States, stronger than physical health, age, education, income, and marital status. Nevertheless, the previous studies do not offer a clear picture of the impacts of education on non-cognitive skills. Heckman et al. (2013) finds positive impacts of education on non-cognitive abilities of three to four years-old children and Dee and West (2011) finds out that reduction in the eighth grade class sizes in Germany improves the non-cognitive skills. Does college education have positive impacts on non-cognitive skills, as Heckman et al. (2013) and Dee and West (2011) find for children in younger ages? Kautz et al. (2014), argue that non-cognitive skills are malleable in adolescent years and they are influenced by schooling. However, Bachman et al. (1978) maintain that college education has little or no impact on self-esteem. Bachman et al. (1978) find that educational attainment does not affect self-esteem when previous levels of self-esteem are controlled for, and Wolfle and Robertshaw (1982) reach an identical finding regarding locus of control. Wolfle and Robertshaw (1982) results show that education beyond high school does not significantly affect locus of control. Nevertheless, Pascarella

and Terenzini (1991) maintain that the locus of control improves during college education years and becomes more internalized, but Wolfle and Robertshaw (1982) suspect that the college students are already internally oriented. They argue that the strongest explanatory variable of locus of control among high school students is ability, such that those with higher abilities have more internalized locus of control. However, Wolfle and List (2004) results show that college education affects the locus of control of the students and shift it from external toward internal levels, even after controlling for ability, background, and previous levels of locus of control.

We apply NLSY79 that contains data from over 12,000 young Americans to study the impacts of college education in the United States on non-cognitive abilities. A variation in the years of schooling that is exogenous to unobservable variables that might influence non-cognitive abilities is essential for overcoming the problem of endogeneity of schooling, since schooling is an endogenous variable. In this paper, we apply a shift in aid policy that affects school attendance of a portion of students, but not all of them to capture exogenous variations in years of schooling. The policy that we apply is the elimination of Social Security student benefits program in 1982, which, from 1965 to 1982, provided aid for the children of Social Security beneficiaries aged 18 to 22.

Note that the focus of the paper is to estimate the impacts of aid eligibility on educational outcomes and non-cognitive abilities rather than the impacts of aid receipt on mentioned variables. Additionally, we focus on eligibility for aid due to the death of father. 90 percent of the students were eligible for the program, because of the death of their fathers. In our two-stage least-square estimations, we have used the interaction of *father deceased* and *before* as an instrument. The interaction of *father deceased* and *before* is an exogenous variable that affects years of education, but does not directly affect non-cognitive abilities. *Father deceased* is a binary variable that indicates whether the father is deceased, and *before* is another binary variable that shows that a student has been a high school senior in the years 1979, 1980, and 1981versus years 1982 and 1983.

Our difference-in-differences results suggest a positive impact of eligibility for student benefits on years of schooling and completing any years of education at college, but these findings should be interpreted with caution due to limitations of NLSY79 dataset, as mentioned before.

The results of the two-stage least-square estimates show that education has a significant impact on non-cognitive skills when the prior levels of non-cognitive skills are not controlled for, but, when the prior levels of non-cognitive skills are controlled for, then we hardly see any significant impacts of education on non-cognitive skills. The results show that the impact of education on self-esteem is significant and positive when a previous level of self-esteem is not included as a control variable, and the impacts are still positive when a previous level of self-esteem is included as a control variable, but in most cases they are not statistically significant.

The results from the two-stage least-square estimations show parallel findings for the impacts of education on locus of control. In most cases, when the previous levels of locus of control are included, the statistically significance level of the coefficients decreases. The coefficients stay positive in all estimations, which suggest that the locus of control shifts from external to internal levels during college years.

Note that the results of this paper do not entirely deny the positive impacts of formal college education on non-cognitive abilities. We also find positive impacts, but when the prior levels of non-cognitive skills are control for, we rarely see any significant impact. Hence, our results are in line with the findings of Wolfle and Robertshaw (1982) and Pascarella and Terenzini (1991), who found similar results decades ago. As mentioned before, several scholars report positive impacts of education on non-cognitive skills, which is a different finding from that of Wolfle and Robertshaw (1982) and Pascarella and Terenzini (1991). Our results do not confirm any causality relationship from college education to non-cognitive skills.

Note that we do not deny the potential positive impacts of the Social Security student aid program on the life and professional outcomes of the beneficiary students. We find positive impacts of the program on education attainment and the program possibly had other positive impacts that could be investigated in the future studies. Additionally, note that this paper does not provide any cost-benefit analysis of the Social Security student aid program. In addition, the sample that was used in this paper, which includes the children of Social Security beneficiaries at high school and college age, might not represent all American students at all levels of education. However, we believe that these results are replicable for samples of students that are more representative of the population of American college students.

Funding: This research received no external funding.

Acknowledgments: I thank Teresa Molina, Sang-Hyop Lee, Timothy Halliday, Nori Tarui, Jennifer Darrah, Michael Roberts, John Lynham, Inessa Love, and Ekaterina Sherstyuk, for valuable inputs and comments. I especially appreciate the guide and support of Andrew Mason. Also, I am thankful of two anonymous reviewers as their comments improved the quality of the paper drastically. I bear sole responsibility for the content of this research and all errors remain mine.

Conflicts of Interest: The author declare no conflict of interest.

## Appendix A. Rosenberg Self-Esteem Scale in NLSY1979

In this appendix, information regarding Rosenberg Self-Esteem Scale in NLSY1979 database has been provided. The following sentence from NLSY1979 and Table A1 show the question that has been asked in the 1987 and 2006 interviews:

"Now I'm going to read a list of opinions people have about themselves. After I read each statement, please tell me how much you strongly agree, agree, disagree or strongly disagree with these opinions".

| Number |  |
|--------|--|
| 1      | I feel that I'm a person of worth, at least on equal basis with others |
| 2      | I feel that I have a number of good qualities                          |
| 3      | All in all, I am inclined to feel that I am a failure                  |
| 4      | I am able to do things as well as most other people                    |
| 5      | I feel I do not have much to be proud of                               |
| 6      | I take a positive attitude toward myself                               |
| 7      | On the whole, I am satisfied with myself                               |
| 8      | I wish I could have more respect for myself                            |
| 9      | I certainly feel useless at times                                      |
| 10     | At times I think I am no good at all                                   |

 Table A1. Rosenberg Self-Esteem Scale questions in NLSY1979 database.

Tables A2–A5 represent data on the respondents of the Rosenberg Self-Esteem Scale questionnaire in NLSY1979. Table A2 shows the number of respondents picked each option from the Rosenberg Self-Esteem Scale questions in a 1987 interview.

| Number of the Question | Strongly Agree | Agree | Disagree | Strongly Disagree | Total  |
|------------------------|----------------|-------|----------|-------------------|--------|
| 1                      | 5780           | 4501  | 155      | 22                | 10,458 |
| 2                      | 5689           | 4674  | 82       | 20                | 10,465 |
| 3                      | 40             | 233   | 4736     | 5450              | 10,459 |
| 4                      | 4807           | 5451  | 150      | 40                | 10,448 |
| 5                      | 127            | 478   | 4696     | 5140              | 10,441 |
| 6                      | 4346           | 5671  | 380      | 46                | 10,443 |
| 7                      | 3313           | 6347  | 729      | 66                | 10,455 |
| 8                      | 299            | 2102  | 5226     | 2812              | 10,439 |
| 9                      | 163            | 2340  | 5236     | 2714              | 10,453 |
| 10                     | 82             | 873   | 5175     | 4328              | 10,458 |

Table A2. Data on Rosenberg Self-Esteem Scale respondents 1987.

Table A3 shows the number of respondents in the 1987 interview, along with their associated Rosenberg Self-Esteem Scale scores.

| Number | of Respondents | Score       |
|--------|----------------|-------------|
|        | 0              | 0           |
|        | 2              | 1 TO 4      |
|        | 4              | 5 TO 9      |
|        | 104            | 10 TO 14    |
|        | 1796           | 15 TO 19    |
|        | 4209           | 20 TO 24    |
|        | 3226           | 25 TO 29    |
|        | 999            | 30 TO 34    |
| -      |                |             |
|        | 10,340         |             |
| Min: 3 | Max: 30        | Mean: 23.37 |

 Table A3. Respondents in 1987 interview along with their associated Rosenberg Self-Esteem Scale scores.

Table A4 shows the number of respondents who picked each option from Rosenberg Self-Esteem Scale questions in 2006 interview.

| Number of the Question | Strongly Agree | Agree | Disagree | Strongly Disagree | Total |
|------------------------|----------------|-------|----------|-------------------|-------|
| 1                      | 4195           | 3225  | 159      | 31                | 7610  |
| 2                      | 4271           | 3285  | 54       | 15                | 7625  |
| 3                      | 44             | 241   | 3538     | 3747              | 7570  |
| 4                      | 3329           | 3865  | 365      | 56                | 7615  |
| 5                      | 119            | 364   | 3498     | 3573              | 7554  |
| 6                      | 3256           | 4032  | 291      | 37                | 7616  |
| 7                      | 2483           | 4408  | 649      | 70                | 7610  |
| 8                      | 219            | 1295  | 3628     | 2386              | 7528  |
| 9                      | 110            | 1261  | 3685     | 2494              | 7550  |
| 10                     | 57             | 485   | 3636     | 3357              | 7535  |

Table A4. Data on Rosenberg Self-Esteem Scale respondents 2006.

Table A5 shows the number of respondents in the 2006 interview, along with their associated Rosenberg Self-Esteem Scale scores.

Table A5. Respondents in 2006 interview along with their associated Rosenberg Self-Esteem Scale scores.

| Number | of Respondents | Score       |
|--------|----------------|-------------|
|        | 0              | 0           |
|        | 0              | 1 TO 4      |
|        | 12             | 5 TO 9      |
|        | 131            | 10 TO 14    |
|        | 1195           | 15 TO 19    |
|        | 2834           | 20 TO 24    |
| 2277   |                | 25 TO 29    |
| 920    |                | 30 TO 34    |
| -      |                |             |
|        | 7369           |             |
| Min: 5 | Max: 30        | Mean: 23.48 |

#### Appendix B. Rotter Locus of Control in NLSY1979

In this appendix, information regarding Rotter Locus of Control Scale in NLSY1979 database has been provided. The following sentence from NLSY1979 and Tables A6 and A7 show the questions that were asked in the 2014 interview:

"I am going to read you four pairs of statements about outlooks on life. For each pair, please select one statement which is closer to your opinion. In addition, tell me whether the statement you select is much closer to your opinion or slightly closer. Try to consider each pair of statements separately when making your choice; do not be influenced by your previous choices".

| Item | Question            |
|------|---------------------|
| 1    | Pair 1, statement A |
| 2    | Pair 1, statement B |
| 3    | Pair 2, statement A |
| 4    | Pair 2, statement B |
| 5    | Pair 3, statement A |
| 6    | Pair 3, statement B |
| 7    | Pair 4, statement A |
| 8    | Pair 4, statement B |

Table A6. Rotter Locus of Control Scale questions in NLSY1979 database.

Table A7 shows the number of respondents who picked each option related to Rotter Locus of Control Scale questions in the 2014 interview. Note that pair 1. B is asked after pair 1. A is asked. Additionally, pair 2. B, pair 3. B, and pair 4. B are asked after pair 2. A, pair 3. A, pair 4. A are asked, respectively.

Pair 1. B to pair 4. B ask a same question which is as follow:

"Is this statement much closer or slightly closer to your opinion?"

| Table A7. | Data on | Rotter | Locus of | Control | respondents 2014. |
|-----------|---------|--------|----------|---------|-------------------|
|-----------|---------|--------|----------|---------|-------------------|

|        | Questio  | Numl   | ber of Responses |             |       |
|--------|--|--|------------------|-------------|-------|
|        | Statement A  | Statement B  | Statement A      | Statement B | Total |
| Pair 1 | What happens to me is my<br>own doing.   | Sometimes I feel that I don't have<br>enough control over the direction my<br>life is taking.                                    | 5784             | 1186        | 6970  |
| Pair 2 | When I make plans, I am almost certain that I can make them work.                        | It is not always wise to plan too far<br>ahead, because many things turn out to<br>be a matter of good or bad<br>fortune anyhow. | 5004             | 2001        | 7005  |
| Pair 3 | In my case, getting what I want has little or nothing to do with luck.                   | Many times, we might just as well decide what to do by flipping a coin.  | 6070             | 835         | 6905  |
| Pair 4 | Many times, I feel that I have little<br>influence over the things that<br>happen to me. | It is impossible for me to believe that<br>chance or luck plays an important role<br>in my life.                                 | 3184             | 3613        | 6797  |

Table A8 shows the number of respondents picked each option from Rotter Locus of Control Scale. Pair B questions in 2014 interview.

Table A8. Data on Locus of Control respondents 2014: pair B.

| Question   | Number of Responses |                    |       |  |  |  |
|------------|---------------------|--------------------|-------|--|--|--|
|            | 1. Much Closer      | 2. Slightly Closer | Total |  |  |  |
| Pair 1. B. | 4863                | 2080               | 6943  |  |  |  |
| Pair 2. B. | 4794                | 2188               | 6982  |  |  |  |
| Pair 3. B. | 4058                | 2826               | 6884  |  |  |  |
| Pair 4. B. | 3184                | 3600               | 6784  |  |  |  |

Table A9 shows the number of respondents in the 2014 interview along with their associated Rotter Locus of Control Scale scores.

| Number of Respondents     |  |  |  |  |
|---------------------------|--|--|--|--|
| 738                       |  |  |  |  |
| 4522                      |  |  |  |  |
| 1316                      |  |  |  |  |
| 39                        |  |  |  |  |
| 6615                      |  |  |  |  |
| Min: 4 Max: 16 Mean: 7.56 |  |  |  |  |
|                           |  |  |  |  |

Table A9. Rotter Locus of Control Scale respondents in 2014 interview along with their associated scores.

#### Appendix C

The findings in Table 3 are from the estimates of the difference-in-differences model. They show the impact of eligibility for Social Security student benefits on education. Several variables are included in the list of the control variables in Table 3. Tables A10–A12 in this appendix provide the results of the estimates of the same model while applying different subsets of control variables that are not provided in Table 3.

Each of Tables A10–A12 contain four columns. The control variables in column (1) of each table include family income, black, Hispanic, father has attended college, mother has attended college, single parent, family size, age in 1988, and female. In column (2), age squared is added to the equation. This shows the sensitivity of the results to the functional form of the regression taken by age. In column (3), we have added four new control variables, including married, which is a binary variable that shows whether a person is married or single. Also, Nonsmsa is included in column (3). Nonsmsa sets to one if individual *i* does not live in a city, and it is zero otherwise. Another covariate in column (3) is Centrsmsa, which equals one if individual *i* lives in the central city. Zasvab is the standardized ASVAB score of individual i, which has been added to NLSY79 dataset in 1979. The ASVAB includes 10 tests that measure skill and knowledge in general science, arithmetic reasoning, paragraph comprehension, word knowledge, coding speed, numerical operations, mathematics knowledge, auto and shop information, electronics information, and mechanical comprehension. In the estimations that are provided in columns (3) and (4), a standardized ASVAB score has been added to our covariates to control for the skills, knowledge, and cognitive abilities of the participants. In column (4), the year of birth fixed effects and region fixed effects are added. The region of birth refers to one of the four regions in the United States: Northeast, the Midwest, the South, and the West.

| Schooling by Age 23             | (1)       | (2)       | (3)       | (4)       |
|---------------------------------|-----------|-----------|-----------|-----------|
| fath on do accord y hofons      | 0.577 **  | 0.572 **  | 0.561 *** | 0.611 **  |
| father deceased $\times$ before | (0.230)   | (0.231)   | (0.234)   | (0.240)   |
| father deceased                 | -0.392 *  | -0.387 *  | -0.422 ** | 0.467 **  |
| lattier deceased                | (0.206)   | (0.207)   | (0.211)   | (0.216)   |
| before                          | 0.584 *** | 0.545 *** | 0.528 *** | 0.657 *** |
|                                 | (0.156)   | (0.163)   | (0.163)   | (0.161)   |
| family income                   | Y         | Y         | Y         | Y         |
| black                           | Y         | Y         | Y         | Y         |
| Hispanic                        | Y         | Y         | Y         | Y         |
| father attended college         | Y         | Y         | Y         | Y         |
| mother attended college         | Y         | Y         | Υ         | Υ         |
| single parent family            | Y         | Y         | Υ         | Υ         |
| family size                     | Y         | Y         | Y         | Y         |
| age88                           | Y         | Y         | Y         | Y         |
| female                          | Y         | Y         | Y         | Y         |
| age^2                           |           | Y         | Y         | Y         |
| married                         |           |           | Y         | Y         |
| nosmsa                          |           |           | Y         | Y         |
| centrsmsa                       |           |           | Y         | Y         |
| Zasvab                          |           |           | Y         | Y         |
| Year of birth fixed effect      |           |           |           | Y         |
| Region fixed effect             |           |           |           | Y         |
| All covariates $\times$ before  |           |           |           |           |
| Number of obs.                  | 886       | 886       | 886       | 875       |
| R-squared                       | 0.118     | 0.119     | 0.147     | 0.157     |

Table A10. Impact of eligibility for Social Security student benefits on years of schooling by age 23.

Note: The dependent variable is years of schooling by age 23. This is the actual number of years of schooling if the respondents in NLSY79 dataset has completed by age 23. The upper numbers in columns (1) to (4) show the coefficient of the independent variable provided in the first column (i.e., column in the left) of the table. The numbers in parentheses represent standard errors. Y means that the associated covariate is included in the estimation of the equation. All regressions are clustered at the household level. \*: Significant at 10%; \*\*: Significant at 5%; \*\*\*: Significant at 1%.

| Complete College by Age 23      | (1)       | (2)       | (3)       | (4)       |
|---------------------------------|-----------|-----------|-----------|-----------|
|                                 | 0.224 *** | 0.224 *** | 0.224 *** | 0.251 *** |
| father deceased $\times$ before | (0.079)   | (0.079)   | (0.080)   | (0.081)   |
| father deceased                 | -0.153 *  | -0.154 ** | -0.167 ** | -0.192 ** |
| rather deceased                 | (0.069)   | (0.069)   | (0.070)   | (0.071)   |
| before                          | 0.135 *** | 0.141 **  | 0.131 **  | 0.140 *** |
| belole                          | (0.050)   | (0.058)   | (0.058)   | (0.058)   |
| family income                   | Y         | Y         | Y         | Y         |
| black                           | Y         | Y         | Y         | Y         |
| Hispanic                        | Y         | Y         | Y         | Y         |
| father attended college         | Y         | Y         | Y         | Y         |
| mother attended college         | Y         | Y         | Y         | Y         |
| single parent family            | Y         | Y         | Y         | Y         |
| family size                     | Y         | Y         | Y         | Y         |
| age88                           | Y         | Y         | Y         | Y         |
| female                          | Y         | Y         | Y         | Y         |
| age^2                           |           | Y         | Y         | Y         |
| married                         |           |           | Y         | Y         |
| nosmsa                          |           |           | Y         | Y         |
| centrsmsa                       |           |           | Y         | Y         |
| Zasvab                          |           |           | Y         | Y         |
| Year of birth fixed effect      |           |           |           | Y         |
| Region fixed effect             |           |           |           | Υ         |
| All covariates × before         |           |           |           |           |
| Number of obs.                  | 895       | 895       | 895       | 884       |

Table A11. Impact of eligibility for Social Security student benefits on complete any years of college by age 23.

Note: The dependent variable is complete any years of college by age 23. This is a binary variable that sets to one if individual *i* has completed any years of college by age 23. It equals to zero otherwise. The upper numbers in columns (1) to (4) show the coefficient of the independent variable provided in the first column (i.e., column in the left) of the table. The numbers in parentheses represent standard errors. Y means that the associated covariate is included in the estimation of the equation. All regressions are clustered at the household level. \*: Significant at 10%; \*\*: Significant at 5%; \*\*\*:

0.093

0.121

0.129

0.093

R-squared

Table A12. Impact of eligibility for Social Security student benefits on complete any years of college by age 28.

| Complete College by Age 28 | (1)       | (2)       | (3)       | (4)       |
|----------------------------|-----------|-----------|-----------|-----------|
| father deceased × before   | 0.316 *** | 0.318 *** | 0.302 *** | 0.325 *** |
| rather deceased × before   | (0.082)   | (0.082)   | (0.082)   | (0.083)   |
| father deceased            | -0.259 *  | -0.261 ** | -0.260 ** | -0.283 ** |
| lattiel deceased           | (0.072)   | (0.072)   | (0.072)   | (0.073)   |
| before                     | 0.121 *** | 0.141 **  | 0.138 *** | 0.144 *** |
| belole                     | (0.050)   | (0.059)   | (0.059)   | (0.061)   |
| family income              | Y         | Y         | Y         | Y         |
| black                      | Y         | Y         | Y         | Y         |
| Hispanic                   | Y         | Y         | Y         | Y         |
| father attended college    | Y         | Y         | Y         | Y         |
| mother attended college    | Y         | Y         | Y         | Y         |
| single parent family       | Y         | Y         | Y         | Y         |
| family size                | Y         | Y         | Y         | Y         |
| age88                      | Y         | Y         | Y         | Y         |
| female                     | Y         | Y         | Y         | Y         |
| age^2                      |           | Y         | Y         | Y         |
| married                    |           |           | Y         | Y         |
| nosmsa                     |           |           | Y         | Y         |
| centrsmsa                  |           |           | Y         | Y         |
| Zasvab                     |           |           | Y         | Y         |
| Year of birth fixed effect |           |           |           | Y         |
| Region fixed effect        |           |           |           | Y         |
| All covariates × before    |           |           |           |           |
| Number of obs.             | 895       | 895       | 895       | 884       |
| R-squared                  | 0.114     | 0.115     | 0.139     | 0.150     |

Note: The dependent variable is complete any years of college by age 28. This is a binary variable that sets to one if individual *i* has completed any years of college by age 28. It equals to zero otherwise. The upper numbers in columns (1) to (4) show the coefficient of the independent variable provided in the first column (i.e., column in the left) of the table. The numbers in parentheses represent standard errors. Y means that the associated covariate is included in the estimation of the equation. All regressions are clustered at the household level. \*: Significant at 10%; \*\*: Significant at 5%; \*\*\*:

#### Appendix D

Tables 6 and 7 show the two-stage least-square estimations of Equation (3). In this appendix, the results of the two-stage least-square estimations are provided while applying different subsets of control variables that are not provided in Tables 6 and 7. In Tables A13–A15, the instrumented variable is schooling by age 23, but in Tables A16–A18 the instrumented variable is complete any years of college by age 23. The interaction of *father deceased* and *before* is applied as our instrument in all regressions.

**Table A13.** Impact of education on Rosenberg Self-Esteem 1987—two-stage least-square estimations.Instrumented variable: schooling by age 23.

| Independent Variable                       | (1)                 | (2)                 | (3)                 | Previous Non-Cognitive<br>Variable is Control Variable |
|--|---------------------|---------------------|---------------------|--|
| Z. Rosenberg 1987                          | 0.287 **<br>(0.113) | 0.304 **<br>(0.117) | 0.260 **<br>(0.118) | NO   |
| Number of obs.                             | 869                 | 869                 | 858                 |  |
| Test of weak instruments:<br>first stage F | 10.784              | 10.037              | 10.919              |  |
| Z. Rosenberg 1987                          | 0.130<br>(0.121)    | 0.148<br>(0.125)    | 0.119<br>(0.125)    | Z. Rosenberg 1980                                      |
| Number of obs.                             | 868                 | 868                 | 857                 |  |
| Test of weak instruments:<br>first stage F | 8.731               | 8.046               | 9.034               |  |

Instrumented: schooling by age 23. Instrument: interaction of *father deceased* and *before*. Test of weak instrument Ho: Instruments are weak. Control variables in column (1) include family income, black, Hispanic, father attended college, mother attended college, single parent family, family size, age, standardized ASVAB score, and female. Column (2) includes all the covariates in column (1) as well as married, nosmsa, and centrsmsa. Column (3) includes all of the covariates in column (2) and year and region fixed effects. All regressions are clustered at the household level. The numbers in parentheses represent standard errors. \*: Significant at 10%; \*\*: Significant at 5%; \*\*\*: Significant at 1%.

**Table A14.** Impact of education on Rosenberg Self-Esteem 2006–two-stage least-square estimations.Instrumented variable: schooling by age 23.

| Independent Variable                       | (1)                 | (2)                 | (3)              | Previous Non-Cognitive<br>Variable is Control Variable |
|--|---------------------|---------------------|------------------|--|
| Z. Rosenberg 2006                          | 0.224 **<br>(0.108) | 0.217 **<br>(0.108) | 0.131<br>(0.116) | NO   |
| Number of obs.                             | 640                 | 640                 | 631              |  |
| Test of weak instruments:<br>first stage F | 9.901               | 9.787               | 11.382           |  |
| Z. Rosenberg 2006                          | 0.161<br>(0.117)    | 0.157<br>(0.117)    | 0.067<br>(0.128) | Z. Rosenberg 1980                                      |
| Number of obs.                             | 639                 | 639                 | 630              |  |
| Test of weak instruments:<br>first stage F | 8.372               | 8.295               | 9.609            |  |

Instrumented: schooling by age 23. Instrument: interaction of *father deceased* and *before*. Test of weak instrument Ho: Instruments are weak. Control variables in column (1) include family income, black, Hispanic, father attended college, mother attended college, single parent family, family size, age, standardized ASVAB score, and female. Column (2) includes all of the covariates in column (1) as well as married, nosmsa, and centrsmsa. Column (3) includes all the covariates in column (2) and year and region fixed effects. All regressions are clustered at the household level. The numbers in parentheses represent standard errors. \*: Significant at 10%; \*\*: Significant at 5%; \*\*\*: Significant at 1%.

| Independent Variable                       | (1)     | (2)     | (3)     | Previous Non-Cognitive<br>Variable is Control Variable |
|--|---------|---------|---------|--|
| Z. Rotter 2014                             | 0.259 * | 0.242   | 0.224   | NO   |
| 2. Rotter 2014                             | (0.155) | (0.151) | (0.137) | 110  |
| Number of obs.                             | 557     | 557     | 549     |  |
| Test of weak instruments:<br>first stage F | 15.246  | 15.136  | 15.301  |  |
| F D 2014                                   | 0.247   | 0.232   | 0.220   | <b>7 D</b> ··· 10 <b>7</b> 0                           |
| Z. Rotter 2014                             | (0.158) | (0.156) | (0.141) | Z. Rotter 1979   |
| Number of obs.                             | 554     | 554     | 546     |  |
| Test of weak instruments:<br>first stage F | 15.520  | 15.342  | 15.087  |  |

**Table A15.** Impact of education on Rotter Locus of Control 2014—two-stage least-square estimations. Instrumented variable: schooling by age 23.

Instrumented: schooling by age 23. Instrument: interaction of *father deceased* and *before*. Test of weak instrument Ho: Instruments are weak. Control variables in column (1) include family income, black, Hispanic, father attended college, mother attended college, single parent family, family size, age, standardized ASVAB score, and female. Column (2) includes all of the covariates in column (1) as well as married, nosmsa, and centrsmsa. Column (3) includes all the covariates in column (2) and year and region fixed effects. All regressions are clustered at the household level. The numbers in parentheses represent standard errors. \*: Significant at 10%; \*\*: Significant at 5%; \*\*\*: Significant at 1%.

**Table A16.** Impact of education on Rosenberg Self-Esteem 1987—two-stage least-square estimations. Instrumented variable: complete any years of college by age 23.

| Independent Variable                       | (1)                 | (2)                 | (3)                 | Previous Non-Cognitive<br>Variable is Control Variable |
|--|---------------------|---------------------|---------------------|--|
| Z. Rosenberg 1987                          | 0.905 **<br>(0.388) | 0.972 **<br>(0.403) | 0.820 **<br>(0.405) | NO   |
| Number of obs.                             | 876                 | 876                 | 865                 |  |
| Test of weak instruments:<br>first stage F | 9.859               | 9.364               | 7.943               | Z. Rosenberg 1980                                      |
| Z. Rosenberg 1987                          | 0.354<br>(0.425)    | 0.418<br>(0.446)    | 0.330<br>(0.444)    |  |
| Number of obs.                             | 875                 | 875                 | 864                 |  |
| Test of weak instruments:<br>first stage F | 7.548               | 7.083               | 6.306               |  |

Instrumented: complete any years of college by age 23. Instrument: interaction of *father deceased* and *before*. Test of weak instrument Ho: Instruments are weak. Control variables in column (1) include family income, black, Hispanic, father attended college, mother attended college, single parent family, family size, age, standardized ASVAB score, and female. Column (2) includes all the covariates in column (1) as well as married, nosmsa, and centrsmsa. Column (3) includes all the covariates in column (2) and year and region fixed effects. All regressions are clustered at the household level. The numbers in parentheses represent standard errors. \*: Significant at 10%; \*\*: Significant at 5%; \*\*\*: Significant at 1%.

**Table A17.** Impact of education on Rosenberg Self-Esteem 2006—two-stage least-square estimations. Instrumented variable: complete any years of college by age 23.

| Independent Variable                       | (1)                 | (2)                 | (3)              | Previous Non-Cognitive<br>Variable is Control Variable |
|--|---------------------|---------------------|------------------|--|
| Z. Rosenberg 2006                          | 0.824 **<br>(0.382) | 0.800 **<br>(0.386) | 0.467<br>(0.356) | NO   |
| Number of obs.                             | 646                 | 646                 | 637              |  |
| Test of weak instruments:<br>first stage F | 8.841               | 8.882               | 9.913            |  |
| Z. Rosenberg 2006                          | 0.588<br>(0.414)    | 0.569<br>(0.419)    | 0.252<br>(0.391) | Z. Rosenberg 1980                                      |
| Number of obs.                             | 645                 | 645                 | 636              |  |
| Test of weak instruments:<br>first stage F | 7.076               | 7.111               | 8.164            |  |

Instrumented: complete any years of college by age 23. Instrument: interaction of *father deceased* and *before*. Test of weak instrument Ho: Instruments are weak. Control variables in column (1) include family income, black, Hispanic, father attended college, mother attended college, single parent family, family size, age, standardized ASVAB score, and female. Column (2) includes all of the covariates in column (1) as well as married, nosmsa, and centrsmsa. Column (3) includes all the covariates in column (2) and year and region fixed effects. All regressions are clustered at the household level. The numbers in parentheses represent standard errors. \*: Significant at 10%; \*\*: Significant at 5%; \*\*\*:

| Independent Variable                       | (1)                | (2)                | (3)              | Previous Non-Cognitive<br>Variable is Control Variable |
|--|--------------------|--------------------|------------------|--|
| Z. Rotter 2014                             | 0.850 *<br>(0.471) | 0.794 *<br>(0.465) | 0.674<br>(0.414) | NO   |
| Number of obs.                             | 562                | 562                | 554              |  |
| Test of weak instruments:<br>first stage F | 11.746             | 12.372             | 12.765           |  |
| Z. Rotter 2014                             | 0.812 *<br>(0.491) | 0.761<br>(0.489)   | 0.660<br>(0.432) | Z. Rotter 1979   |
| Number of obs.                             | 559                | 559                | 551              |  |
| Test of weak instruments:<br>first stage F | 11.206             | 11.807             | 12.489           |  |

**Table A18.** Impact of education Rotter Locus of Control 2014—two-stage least-square estimations. Instrumented variable: complete any years of college by age 23.

Instrumented: complete any years of college by age 23. Instrument: interaction of *father deceased* and *before*. Test of weak instrument Ho: Instruments are weak. Control variables in column (1) include family income, black, Hispanic, father attended college, mother attended college, single parent family, family size, age, standardized ASVAB score, and female. Column (2) includes all the covariates in column (1) as well as married, nosmsa, and centrsmsa. Column (3) includes all the covariates in column (2) and year and region fixed effects. All regressions are clustered at the household level. The numbers in parentheses represent standard errors. \*: Significant at 10%; \*\*: Significant at 5%; \*\*\*: Significant at 1%.

#### References

- Arcidiacono, Peter, Patrick Bayer, and Aurel Hizmo. 2010. Beyond signaling and human capital: Education and the revelation of ability. *American Economic Journal: Applied Economics* 2: 76–104. [CrossRef]
- Bachman, Jerald G., Patrick M. O'Malley, and Jerome Johnston. 1978. *Adolescence to Adulthood—Change and Stability in the Lives of Young Men.* Washington, DC: Institute for Social Research, University of Michigan.
- Baumeister, Roy F., Jennifer D. Campbell, Joachim I. Krueger, and Kathleen D. Vohs. 2003. Does high self-esteem cause better performance, interpersonal success, happiness, or healthier lifestyles? *Psychological Science in the Public Interest* 4: 1–44. [CrossRef] [PubMed]
- Becker, Gary S. 1964. Human capital. In *A Theoretical and Empirical Analysis with Special References to Education*. New York: National Bureau of Economic Research, distributed by Columbia University Press.
- Becker, Gary S. 1992. Human capital and the economy. Proceedings of the American Philosophical Society 136: 85–92.
- Ben-Porath, Yoram. 1967. The production of human capital and the life cycle of earnings. *Journal of Political Economy* 75: 352–65. [CrossRef]
- Blair, Clancy, David Gamson, Steven Thorne, and David Baker. 2005. Rising mean IQ: Cognitive demand of mathematics education for young children, population exposure to formal schooling, and the neurobiology of the prefrontal cortex. *Intelligence* 33: 93–106. [CrossRef]
- Boden, Joseph M., David M. Fergusson, and L. John Horwood. 2007. Self-esteem and violence: Testing links between adolescent self-esteem and later hostility and violent behavior. *Social Psychiatry and Psychiatric Epidemiology* 42: 881–91. [CrossRef]
- Boden, Joseph M., David M. Fergusson, and L. John Horwood. 2008. Does adolescent self-esteem predict later life outcomes? A test of the causal role of self-esteem. *Development and Psychopathology* 20: 319–39. [CrossRef]
- Bouchard, Thomas J., and John C. Loehlin. 2001. Genes, evolution, and personality. *Behavior Genetics* 31: 243–73. [CrossRef]
- Bowles, Terry. 1999. Focusing on time orientation to explain adolescent selfconcept and academic achievement. *Journal of Applied Health Behavior* 1: 1–8.
- Bowles, Samuel, Herbert Gintis, and Melissa Osborne. 2001. The determinants of earnings: A behavioral approach. *Journal of Economic Literature* 39: 1137–76. [CrossRef]
- Brickman, Phillip, and Ronnie J. Bulman. 1977. Pleasure and pain in social comparison. *Social Comparison Processes: Theoretical and Empirical Perspectives* 149: 186.
- Butler, Ruth. 1989. Mastery versus ability appraisal: A developmental study of children's observations of peers' work. *Child Development* 60: 1350–61. [CrossRef]

- Caliendo, Marco, Deborah A. Cobb-Clark, and Arne Uhlendorff. 2015. Locus of control and job search strategies. *Review of Economics and Statistics* 97: 88–103. [CrossRef]
- Carlsson, Magnus, Gordon B. Dahl, Björn Öckert, and Dan-Olof Rooth. 2015. The effect of schooling on cognitive skills. *Review of Economics and Statistics* 97: 533–47. [CrossRef]
- Cascio, Elizabeth U., and Ethan G. Lewis. 2006. Schooling and the armed forces qualifying test evidence from school-entry laws. *Journal of Human resources* 41: 294–318. [CrossRef]
- Casner-Lotto, Jill, and Linda Barrington. 2006. Are they Really Ready to Work? Employers' Perspectives on the Basic Knowledge and Applied Skills of New Entrants to the 21st Century US Workforce. Partnership for 21st Century Skills. Washington: The Conference Board.
- Ceci, Stephen J. 1991. How much does schooling influence general intelligence and its cognitive components? A reassessment of the evidence. *Developmental Psychology* 27: 703. [CrossRef]
- Cliffordson, Christina, and Jan-Eric Gustafsson. 2008. Effects of age and schooling on intellectual performance: Estimates obtained from analysis of continuous variation in age and length of schooling. *Intelligence* 36: 143–52. [CrossRef]
- Cobb-Clark, Deborah A. 2015. Locus of control and the labor market. IZA Journal of Labor Economics 4: 3. [CrossRef]

Crocker, Jennifer, and Connie T. Wolfe. 2001. Contingencies of self-worth. Psychological Review 108: 593. [CrossRef]

- Crocker, Jennifer, Samuel R. Sommers, and Riia K. Luhtanen. 2002. Hopes dashed and dreams fulfilled: Contingencies of self-worth and graduate school admissions. *Personality and Social Psychology Bulletin* 28: 1275–86. [CrossRef]
- de Araujo, Pedro, and Stephen Lagos. 2013. Self-esteem, education, and wages revisited. *Journal of Economic Psychology* 34: 120–32. [CrossRef]
- Dee, Thomas S., and Martin R. West. 2011. The non-cognitive returns to class size. *Educational Evaluation and Policy Analysis* 33: 23–46. [CrossRef]
- Deke, John, and Joshua Haimson. 2006. Valuing Student Competencies: Which Ones Predict Postsecondary Educational Attainment and Earnings, and for Whom? Final Report. Princeton: Mathematica Policy Research, Inc.
- Diener, Ed. 1984. Subjective well-being. *Psychological Bulletin* 95: 542–75. [CrossRef]
- Drago, Francesco. 2011. Self-esteem and earnings. Journal of Economic Psychology 32: 480-88. [CrossRef]
- Dynarski, Susan M. 2003. Does aid matter? Measuring the effect of student aid on college attendance and completion. *American Economic Review* 93: 279–88. [CrossRef]
- Fortin, Nicole M. 2008. The gender wage gap among young adults in the united states the importance of money versus people. *Journal of Human Resources* 43: 884–918. [CrossRef]
- Garcia, Emma. 2016. The need to address non-cognitive skills in the education policy agenda. In *Non-Cognitive Skills and Factors in Educational Attainment*. Washington, DC: Brill Sense, pp. 31–64.
- Gintis, Herbert. 1971. Education, technology, and the characteristics of worker productivity. *The American Economic Review* 61: 266–79.
- Gustafsson, Jan-Eric. 2016. Lasting effects of quality of schooling: Evidence from PISA and PIAAC. *Intelligence* 57: 66–72. [CrossRef]
- Hansen, Karsten T., James J. Heckman, and Kathleen J. Mullen. 2004. The effect of schooling and ability on achievement test scores. *Journal of econometrics* 121: 39–98. [CrossRef]
- Harvey, J. Michael. 1971. Locus of control shift in administrators. Perceptual and Motor Skills 33: 980-82. [CrossRef]
- Heckman, James J., and Tim Kautz. 2013. *Fostering and Measuring Skills: Interventions that Improve Character and Cognition*. (No. w19656). Cambridge: National Bureau of Economic Research.
- Heckman, James J., and Yona Rubinstein. 2001. The importance of noncognitive skills: Lessons from the GED testing program. *American Economic Review* 91: 145–49. [CrossRef]
- Heckman, James J., Jora Stixrud, and Sergio Urzua. 2006. The effects of cognitive and noncognitive abilities on labor market outcomes and social behavior. *Journal of Labor Economics* 24: 411–82. [CrossRef]
- Heckman, James J., Rodrigo Pinto, and Peter Savelyev. 2013. Understanding the mechanisms through which an influential early childhood program boosted adult outcomes. *American Economic Review* 103: 2052–86. [CrossRef] [PubMed]
- Heineck, Guido, and Silke Anger. 2010. The returns to cognitive abilities and personality traits in Germany. *Labour Economics* 17: 535–46. [CrossRef]
- Murray, Charles, and Richard Herrnstein. 1994. The Bell Curve: The Reshaping of American Life by Differences in Intelligence. New York: Free Press.

- Jonsson, Bert, Maria Waling, Anna S. Olafsdottir, Hanna Lagström, Hege Wergedahl, Cecilia Olsson, Eldbjørg Fossgard, Asle Holthe, Sanna Talvia, Ingibjorg Gunnarsdottir, and et al. 2017. The effect of schooling on basic cognition in selected Nordic Countries. *Europe's Journal of Psychology* 13: 645. [CrossRef]
- Kautz, Tim, James J. Heckman, Ron Diris, Bas Ter Weel, and Lex Borghans. 2014. Improving Cognitive and Non-Cognitive Skills to Promote Lifetime Success. (No. w20749). Cambridge: National Bureau of Economic Research.
- Kuhn, Peter, and Catherine Weinberger. 2005. Leadership skills and wages. *Journal of Labor Economics* 23: 395–436. [CrossRef]
- Kuster, Farah, Ulrich Orth, and Laurenz L. Meier. 2013. High self-esteem prospectively predicts better work conditions and outcomes. *Social Psychological and Personality Science* 4: 668–75. [CrossRef]
- Marshall, Sarah L., Phillip D. Parker, Joseph Ciarrochi, and Patrick CL Heaven. 2014. Is self-esteem a cause or consequence of social support? A 4-year longitudinal study. *Child Development* 85: 1275–91. [CrossRef]
- Murnane, Richard J., John B. Willett, M. Jay Braatz, and Yves Duhaldeborde. 2001. Do different dimensions of male high school students' skills predict labor market success a decade later? Evidence from the NLSY. *Economics of Education Review* 20: 311–20. [CrossRef]
- Neal, Derek A., and William R. Johnson. 1996. The role of premarket factors in black-white wage differences. *Journal of Political Economy* 104: 869–95. [CrossRef]
- Orth, Ulrich, and Richard W. Robins. 2014. The development of self-esteem. *Current Directions in Psychological Science* 23: 381–87. [CrossRef]
- Orth, Ulrich, Richard W. Robins, Kali H. Trzesniewski, Jürgen Maes, and Manfred Schmitt. 2009. Low self-esteem is a risk factor for depressive symptoms from young adulthood to old age. *Journal of Abnormal Psychology* 118: 472. [CrossRef] [PubMed]
- Orth, Ulrich, Richard W. Robins, and Keith F. Widaman. 2012. Life-span development of self-esteem and its effects on important life outcomes. *Journal of Personality and Social Psychology* 102: 1271. [CrossRef] [PubMed]
- Pascarella, Ernest T., and Patrick T. Terenzini. 1991. *How College Affects Students: Findings and Insights from Twenty Years of Research*. San Francisco: Jossey-Bass Inc.
- Pomerantz, Eva M., Diane N. Ruble, Karin S. Frey, and Faith Greulich. 1995. Meeting goals and confronting conflict: Children's changing perceptions of social comparison. *Child Development* 66: 723–38. [CrossRef]
- Rosenberg, Morris. 1965. Society and the Adolescent Self-image. Princeton: Princeton University Press.
- Rosenberg, Morris, Carmi Schooler, and Carrie Schoenbach. 1989. Self-esteem and adolescent problems: Modeling reciprocal effects. *American Sociological Review* 54: 1004–18.
- Rotter, Julian. B. 1966. Generalized expectancies for internal versus external control of reinforcement. *Psychological Monographs: General and Applied* 80: 1–28. [CrossRef]
- Rubin, Rosalyn A., Jeanne Dorle, and Susanne Sandidge. 1977. Self-esteem and school performance. *Psychology in the Schools* 14: 503–507. [CrossRef]
- Ruble, Diane N., and Karin S. Frey. 1991. Changing Patterns of Comparative Behavior as Skills are Acquired: A Functional Model of Self-Evaluation. In *Social Comparison: Contemporary Theory and Research*. Edited by Jerry Suls and Thomas Ashby Ed Wills. Hillsdale: Lawrence Erlbaum Associates, Inc., pp. 79–113.
- Skaalvik, Einar M., and Knut A. Hagtvet. 1990. Academic achievement and self-concept: An analysis of causal predominance in a developmental perspective. *Journal of Personality and Social Psychology* 58: 292. [CrossRef]
- Spence, Michael. 1978. Job market signaling. In Uncertainty in Economics. Cambridge: Academic Press, pp. 281–306.
- Stelzl, Ingeborg, Ferdinand Merz, Theodor Ehlers, and Herbert Remer. 1995. The effect of schooling on the development of fluid and cristallized intelligence: A quasi-experimental study. *Intelligence* 21: 279–96. [CrossRef]
- Taylor, Shelley E., and Marci Lobel. 1989. Social comparison activity under threat: Downward evaluation and upward contacts. *Psychological Review* 96: 569. [CrossRef]
- Trzesniewski, Kali H., M. Brent Donnellan, Terrie E. Moffitt, Richard W. Robins, Richie Poulton, and Avshalom Caspi. 2006. Low self-esteem during adolescence predicts poor health, criminal behavior, and limited economic prospects during adulthood. *Developmental Psychology* 42: 381. [CrossRef]
- Waddell, Glen R. 2006. Labor-Market Consequences of Poor Attitude and Low Self-Esteem in Youth. *Economic Inquiry* 44: 69–97. [CrossRef]
- Winship, Christopher, and Sanders Korenman. 1997. Does staying in school make you smarter? The effect of education on IQ in The Bell Curve. In *Intelligence, Genes, and Success*. New York: Springer, pp. 215–34.

27 of 27

- Wolfle, Lee M., and Jill H. List. 2004. Temporal stability in the effects of college attendance on locus of control, 1972–1992. *Structural Equation Modeling* 11: 244–60. [CrossRef]
- Wolfle, Lee M., and Dianne Robertshaw. 1982. Effects of college attendance on locus of control. *Journal of Personality* and Social Psychology 43: 802. [CrossRef]



© 2020 by the author. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).